Author's response to reviews

Title: Central trigeminocardiac reflex in pediatric neurosurgery: First description by a case report

Authors:

Toma Y Spiriev (spiriev@gmail.com)
Christo Tz Tzekov (tzekovchr@abv.bg)
Lili N Laleva (lililaleva@gmail.com)
Christina Kostadinova (hris.kostadinova@gmail.com)
Slavomir I Kondoff (slavi.kondoff@gmail.com)
Nora Sandu (nsandu@gmail.com)
Bernhard J Schaller (bernhardjschaller@gmail.com)

Version: 2 Date: 28 June 2012

Author's response to reviews:

Dear editor,

We have changed the manuscript according to the revisions required:
- we have included the patient informed consent section
- we have included removed the names of the hospital, department
- we have changed the disclosure section according to the requirements

We would like to thank you for your attention and consideration and we hope that these changes would be sufficient.

Yours Sincerely,

Toma Spiriev, MD
Department of Neurosurgery,
Tokuda Hospital Sofia,
Bulgaria